 Vc/8-Feb-05
Week 6 starting 07 Feb 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 31 Jan 05 @  2/7/2005 3:24:39 PM    

	
	A1
	MBAL2123
	Prep 5 Pumping + Cool Down
	

	
	A2
	MBAR3284
	ICS 1 WP04 HV Test Warm
	

	
	B1
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	B2
	MBBL1193
	ICS 2 Connect Magnet (ICS)
	

	
	C1
	MBAR3281
	PT 12 Warm Up
	

	
	C2
	MBAR3253
	Current Lead Upgrade
	

	
	D1
	SSS127
	ICS 4 Final connection
	

	
	D2
	SSS098
	PT 13.1 IAP @ Warm
	

	
	E1
	MBAR3282
	ICS 5 Traveller Last
	

	
	E2
	MBBR3215
	ICS 4 Final connection
	

	
	F1
	MBBR2525
	PT 1 IAP @ cold
	

	
	F2
	MBAL3104
	Prep 5 Pumping + Cool Down
	


Coordinator this week: P Pugnat; MTM support on call :   16-3602 .

144  Magnets Tested  ( SSS036 twice)
10  dipoles( 0 with MM & long MM)+ 4  SSS(0  with MM) +  0  SSSS 
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

( 2123 had busbar quenches; 3278 thermal cycle )
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
· Silver Shoes bench F etc
General Points 
Problems   Power Converters Software issue & reset

Silver Shoe repairs gong on still  ( outstanding still on C2, F cluster)

SSS36 had Insulation Issues?

Cryo Level problems so 2123 quenching in busbars & so was disconnected & will come back again on A bench 

Power Converters software bug appeared Tues meaning needing reset before every use ( OP complained like hell) no solution till Fri pm via PC routine

Sun aft to Mon morning, there were no MRB prepared magnets ( all 3 already used on Sat ) & so bench had to be kept empty

AOB

	


Tests 

	Magnets under Test from 30-01-2005 to 05-02-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAL1173 
	[image: image3.png]



	29-01-2005 
	28 h.
	79 h.
	15 h.
	13 h.
	11 h.
	04-02-2005 
	144 hours
	
	

	2
	TBA2
	MBAR3278 
	[image: image4.png]



	28-01-2005 
	35 h.
	29 h.
	66 h.
	18 h.
	0 h.
	03-02-2005 
	144 hours
	
	

	3
	TBB1
	SSS056 
	[image: image5.png]



	24-01-2005 
	43 h.
	8 h.
	19 h.
	8 h.
	17 h.
	31-01-2005 
	168 hours
	
	

	4
	TBB2
	MBBR3271 
	[image: image6.png]



	28-01-2005 
	27 h.
	25 h.
	27 h.
	16 h.
	12 h.
	02-02-2005 
	120 hours
	
	

	5
	TBD1
	SSS036 
	[image: image7.png]



	20-01-2005 
	44 h.
	34 h.
	17 h.
	17 h.
	59 h.
	31-01-2005 
	264 hours
	
	

	6
	TBD1
	SSS036 
	[image: image8.png]



	31-01-2005 
	57 h.
	16 h.
	27 h.
	19 h.
	0 h.
	05-02-2005 
	120 hours
	
	

	7
	TBD2
	SSS058 
	[image: image9.png]



	28-01-2005 
	56 h.
	20 h.
	40 h.
	6 h.
	9 h.
	03-02-2005 
	144 hours
	
	

	8
	TBE1
	MBAR3280 
	[image: image10.png]



	30-01-2005 
	28 h.
	25 h.
	23 h.
	15 h.
	8 h.
	03-02-2005 
	96 hours
	
	

	9
	TBE1
	MBBL1145 
	[image: image11.png]



	24-01-2005 
	101 h.
	54 h.
	20 h.
	16 h.
	10 h.
	30-01-2005 
	144 hours
	
	

	10
	TBE2
	MBAL3108 
	[image: image12.png]



	29-01-2005 
	22 h.
	29 h.
	15 h.
	15 h.
	4 h.
	02-02-2005 
	96 hours
	
	

	11
	TBF1
	MBAL2128 
	[image: image13.png]



	24-01-2005 
	74 h.
	30 h.
	26 h.
	18 h.
	20 h.
	31-01-2005 
	168 hours
	
	

	12
	TBF1
	MBAR3283 
	[image: image14.png]



	31-01-2005 
	26 h.
	22 h.
	33 h.
	14 h.
	8 h.
	04-02-2005 
	96 hours
	
	

	13
	TBF2
	MBAL1174 
	[image: image15.png]



	01-02-2005 
	30 h.
	23 h.
	33 h.
	13 h.
	15 h.
	05-02-2005 
	96 hours
	
	

	14
	TBF2
	MBAL2123 
	[image: image16.png]



	24-01-2005 
	36 h.
	27 h.
	30 h.
	19 h.
	56 h.
	01-02-2005 
	192 hours
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL1173
	12068 A.
(D1_U)
	12454.8 A.
(D2_U)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR3278
	9740 A.
(D1_U)
	11736 A.
(D2_U-S01)
	11808 A.
(D2_U_S01)
	12108 A.
(D2-U-S01)
	10813 A.
(D1-U)
	7197.4 A.
(D1-U-S02)
	12401.3 A.
(D2_U/S01)
	12724 A.
(D2_U_S01)
	12850 A.
[image: image18.jpg]



	-
	-
	-
	-

	3
	SSS056
	12612.6 A.
(QI_12)
	0 A.
(No Data)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBR3271
	11263.7 A.
(D2-U)
	12426 A.
(D1- U)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS036
	[image: image21.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS036
	[image: image22.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS058
	11860 A.
(QE_34)
	12125.3 A.
(QE-34)
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR3280
	11542.5 A.
(D2-U)
	12655 A.
(D2_L)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL1145
	11985.5 A.
(D1_U)
	12300.1 A.
(D1-L)
	[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAL3108
	11162 A.
(D1-U)
	12685 A.
(D2_U)
	[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2128
	11108 A.
(D1_L)
	12850 A.
(D1-U)
	[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR3283
	8920.1 A.
(D2-L)
	9136 A.
(No Data)
	11010.6 A.
(D2_L)
	12322 A.
(D2_U)
	12809 A.
(D2-L)
	12850 A.
[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-

	13
	MBAL1174
	11927 A.
(D1-U)
	12746 A.
(D2_L)
	[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAL2123
	6245.7 A.
(Vbus-Int)
	6185.3 A.
(Vbus-Int)
	6110.6 A.
(Vbus-Int)
	[image: image30.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


	From 30-01-2005 to 05-02-2005 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBAL1174 
	False 
	01-02-2005 
	30 h.
	23 h.
	33 h.
	13 h.
	15 h.
	05-02-2005 
	4 days
	2 

	2
	TBF2
	MBAL2123 
	False 
	24-01-2005 
	36 h.
	27 h.
	30 h.
	19 h.
	56 h.
	01-02-2005 
	8 days
	3 

	3
	TBF1
	MBAL2128 
	False 
	24-01-2005 
	74 h.
	30 h.
	26 h.
	18 h.
	20 h.
	31-01-2005 
	7 days
	2 

	4
	TBF1
	MBAR3283 
	False 
	31-01-2005 
	26 h.
	22 h.
	33 h.
	14 h.
	8 h.
	04-02-2005 
	4 days
	5 

	5
	TBE2
	MBAL3108 
	False 
	29-01-2005 
	22 h.
	29 h.
	15 h.
	15 h.
	4 h.
	02-02-2005 
	4 days
	2 

	6
	TBE1
	MBAR3280 
	False 
	30-01-2005 
	28 h.
	25 h.
	23 h.
	15 h.
	8 h.
	03-02-2005 
	4 days
	2 

	7
	TBE1
	MBBL1145 
	False 
	24-01-2005 
	101 h.
	54 h.
	20 h.
	16 h.
	10 h.
	30-01-2005 
	6 days
	2 

	8
	TBD2
	SSS058 
	False 
	28-01-2005 
	56 h.
	20 h.
	40 h.
	6 h.
	9 h.
	03-02-2005 
	6 days
	2 

	9
	TBD1
	SSS036 
	False 
	20-01-2005 
	44 h.
	34 h.
	17 h.
	17 h.
	59 h.
	31-01-2005 
	11 days
	0 

	10
	TBD1
	SSS036 
	False 
	31-01-2005 
	57 h.
	16 h.
	27 h.
	19 h.
	0 h.
	05-02-2005 
	5 days
	0 

	11
	TBB2
	MBBR3271 
	False 
	28-01-2005 
	27 h.
	25 h.
	27 h.
	16 h.
	12 h.
	02-02-2005 
	5 days
	2 

	12
	TBB1
	SSS056 
	False 
	24-01-2005 
	43 h.
	8 h.
	19 h.
	8 h.
	17 h.
	31-01-2005 
	7 days
	2 

	13
	TBA2
	MBAR3278 
	True 
	28-01-2005 
	35 h.
	29 h.
	66 h.
	18 h.
	0 h.
	03-02-2005 
	6 days
	8 

	14
	TBA1
	MBAL1173 
	False 
	29-01-2005 
	28 h.
	79 h.
	15 h.
	13 h.
	11 h.
	04-02-2005 
	6 days
	2 


Week 30Jan05 to 5 Feb 05
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Silver Shoe repairs gong on still  ( outstanding still on C2, F cluster)
SSS36 had Insulation Issues?

Cryo Level problems so 2123 quenching in busbars & so was disconnected & will come back again on A bench 

Power Converters software bug appeared Tues meaning needing reset before every use ( OP complained like hell) no solution till Fri pm via PC routine

Sun aft to Mon morning, there were no MRB prepared magnets ( all 3 already used on Sat ) & so bench had to be kept empty

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _01_M


	TBA2
	Noisy signal current during PTE
	AB/PO
	
	Consequence of the software change.

Giorgio contacted Q.King. He advised to perform a reset on RMA (each time befor a 3 kA?)

How do we do this? Ne wfacilyti installed on the PCLHC754006\46 next to cluster "C" console.

	Tuesday _01_M


	TBF2
	Hf,lf DAQ problem 
	AB/CO
	Hf,lf
	Salem  informed A.Raimondo Problem in CPU card swapped from TBC to TBF.

	Tuesday _01_A


	TBF2
	SSS058; Cannot power the EF circuit in TBF.


	MTM
	
	Investigation by Giorgio. Problem in the clamp. NC opened.

	Tuesday _01_N


	TBD2
	Cryo problem: cool down to 1.9 K stopped.


	ACR
	
	I

	Tuesday _01_N


	TBD1
	IAP@warm: PLCSE fault not possible to reset


	MTM
	
	

	Tuesday _01_N

Wednesday _02_M
	TBA2
	SPA abort during LHC cycle

Again during training quench.
	?
	
	Everything OK.

	Wednesday _02_A


	TBF1
	IDCCT not connected: no recording of the current for LF QH test.


	MTM
	
	Reconnected and test repeated

	Thursday _03_M


	TBF1
	Quench in Vbus int at 9 kA.


	MTM/ACR
	
	Electronic checked by Giorgio. OK.

Test repeated but He Liq level in the current leads of 75% (60% is enough normally).Training quench performed.



	Thursday _03_A


	TBD1
	Cooling down below 5 K difficult.

He level problem?


	ACR
	
	2 h of stop

	Thursday _03_A


	TBC1
	IAP test @ warm. Files missing : unable to launch the tests. Retry: application hung.


	AB/CO
	IAP
	Adrian contacted. He solved the problem

	Thursday _03_A


	TBF1
	Hf DAQ did not rearm itself. After a training quench. 
	AB/CO
	hf
	Salem contacted Adrian.

	Thursday _03_N


	TBF2
	IAP @ warm: permanent triggers on Vbus Vsupra cards  


	MTM
	
	VT2 and VT2_5 opened. Salem by passed the 2 Taps in the crates

	Friday _04_M


	TBF2
	Alarm on QH2: TminV<500 ms.  


	MTM
	AQA
	Salem called. Asked to repeat. Same

He advised to proceed.

	Friday _04_M


	TBD2
	HV test @warm failed


	MTM
	
	Salem and Andre investigated. Performed with the switch in the non correct position.

	Friday _04_M
	TBA2
	Compressor stopped 
	ACR
	
	2h of stops.



	Saturday_05_M


	TBE1
	Wrong name (3245 instead of 3282) of the magnet for the HV test


	MTM
	HV
	File corrected manually

	Saturday_05_N


	TBC1
	LF_Web gives a vi error when starting-up. LF and HF status windows do not display anything.
	AB-CO
	lf
	Salem contacted. He will inform Adrian Monday

	Saturday_05_N


	TBD2
	SSS098: IAP at Cold (A.C compensation) amplitude error for QI 12-34,QE12-34,QI-E. 
Offsets could not be adjusted. 
Vbus Int and Ext trigger when the source voltage exceeds 2.6V. 
Quench at 15 A in Vbus int
	MTM
	
	Salem found the origin : adaptator (for right SSS) to match with the VT configuration was not present.

Same origin.

	Saturday_05_N


	TBA2
	 IAP at warm (13.1) could not be launched, as triggers are present on 'V supra_ext_Lead and Vclamp_Ext' .


	MTM
	
	Salem removed a by pass on clamp V_taps for Test on A1. I put it back for A2, then OK

	Sunday_06_M


	TBA1
	Short anticryostat resistance in aperture 2 opened. 


	MTM
	
	Salem checked the T value reading . Can survive with the two others. He authorized to go on.

	Sunday_06_A


	TBD2
	Powering of the CD, EF circuits not possible. 


	ICS
	
	Salem found that the connection is not done in the proper way (inverted). He powered the correctors manually.

	Monday_07_M


	TBF1
	MM1 on 2525 magnet : rotation problem in aperture 2


	AB/CO
	
	ICS called mechanic team.

	Monday_07_M


	TBC1
	De-excitation test launched without hf DAQ because it cannot be armed


	AB/CO
	
	Salem changed the card 3. Test repeated.


 










